

Received November 3, 2019, accepted November 12, 2019, date of publication November 27, 2019, date of current version December 13, 2019.

Digital Object Identifier 10.1109/ACCESS.2019.2955142

# **Congestion Avoidance in Vehicular Networks: A Contemporary Survey**

BALAWAL SHABIR<sup>®1</sup>, MUAZZAM A. KHAN<sup>®1</sup>, ANIS U. RAHMAN<sup>®1,2</sup>, ASAD W. MALIK<sup>®1,2</sup>, AND ABDUL WAHID<sup>®1</sup>

<sup>1</sup>National University of Sciences and Technology (NUST), Islamabad 44000, Pakistan

<sup>2</sup>Department of Information Systems, Faculty of Computer Science and Information Technology, University of Malaya, Kuala Lumpur 50603, Malaysia Corresponding author: Muazzam A. Khan (muazzam,khattak@seecs.edu.pk)

**ABSTRACT** Congestion in vehicular ad hoc networks affects the performance of delay-sensitive applications when exchanging emergency or general information sharing messages. In particular, during emergencies on roads like road accidents and security warnings demand high reliability and low latency. However, in traditional solutions, such messages use the same control channel for transmission leading up to a saturated or congested channel. Furthermore, the highly dynamic nature of vehicular networks leading up to unpredictable routing patterns, subsequently degrading the network performance. An uneven deliverance of these critical messages can be catastrophic for the delay-sensitive applications. Thus, congestion control remains one of the most challenging problems within the domain of vehicular networks. This paper provides a comprehensive overview of the working of vehicular networks, the recent research advances to cater for congestion in such networks, and open problems and challenges relevant to congestion avoidance.

**INDEX TERMS** Ad hoc vehicular network, congestion avoidance, power-based, rate-based, priority-based, clustering-based, CSMA/CA-based, hybrid, vehicular simulation.

NOMENCLAT	URE	ECPR	environmental and context-aware combined			
ACRONYMS			power and rate distributed congestion			
BEB	binary exponential back-off		control			
BPR	Bayesian prediction ranking	EDF	earliest deadline first			
C-ITS	cooperative intelligent transport system	EEADP	effective and efficient adaptive probabilistic			
CAM	cooperative awareness messages		data dissemination protocol			
CBR	channel busy ratio	GloMoSim	global mobile information system simulator			
CBT	channel busy time	IDP	inter packet delay			
CCW	cooperative collision warning	IRT	inter-reception time			
CPB	clustering-based probabilistic broadcasting	JiST	Java in simulation time			
CPRC	combined power and rate control	LCA	lane change assistance linear message rate integrated control			
CRN	congestion road notification	LIMERIC				
CSMA/CA	carrier-sense multiple access with collision	MoNoTrac	mobile node trace generator			
	avoidance	MOVE	mobility model generator for vehicular net-			
DBSMA	dynamic broadcast storm mitigation		works			
	approach	MPC	multi-metric tx-power control protocol object-oriented tool command language objective modular network test-bed in C++ optimized network engineering tools			
DCF	distribution coordination function	OTCL				
DSRC	dedicated short range communication	OMNeT++				
DCP-ABE	Distributed cipher policy attribute based	OPNET				
	encryption	PCW	predictive contention window			
		PDR	packet delivery ratio			
		PE-MAC	power efficient media access control			
The associate	e editor coordinating the review of this manuscript and	PLE	path loss exponent			
approving it for p	ublication was Moayad Aloqaily <sup>10</sup> .	SBAPC	speed-based adaptive power control			

173196 This work is licensed under a Creative Commons Attribution 4.0 License. For more information, see http://creativecommons.org/licenses/by/4.0/ VOLUME 7, 2019

SDSCC	static and dynamic scheduling for conges-
	tion control
SIR	signal-to-interference ratio
SR-CSMA	safety range-carrier sense multiple access
SFIR	SDN and fog-based intersection routing
SESAC	SDN-Enabled Social-Aware Clustering
STRAW	street random waypoint
SUMO	simulation of urban mobility
TMS	traffic management system
TPM	trajectory predicted mechanism
TPRC	joint transmission power rate control
TTC	time to collision
U-EDF	uni-priority safety messages message dis-
	semination using earliest deadline first
UFC	unified framework of clustering
VDBPC	vehicle density-based power control
VENTOS	vehicular network open simulator
VMaSC-LTE	vehicular multihop algorithm for stable
	clustering - Long term evolution
VSIMRTI	V2X simulation runtime infrastructure
XML	extensible markup language

## I. INTRODUCTION

Vehicular ad hoc networks (VANETs) are an active research area since it offers a wide variety of services and applications. The essential services and features include enhanced traffic efficiency, infotainment, and passenger safety. VANETs make the journey comfortable for passengers and drivers by performing efficient traffic management and road safety. VANETs primarily have two modes of operations, i.e. vehicle-to-vehicle (V2V) and vehicle-to-infrastructure (V2I). In V2V, vehicles directly communicate in a distributed fashion whereas in V2I mode vehicles communicate through the roadside unit (RSU) as illustrated in Figure 1. The vehicles in VANET architecture usually send and share two types of messages: safety general and safety-critical messages [1]. The non-safety or safety general messages include infotainment services which are usually delayed tolerant, whereas, safety general messages are delay-insensitive. A dedicated short-range communication (DSRC) at 5.9GHZ is provided by a standardized IEEE 802.11p WAVE (wireless access in vehicular environments) protocol.

Network congestion is a severe issue in VANETs. Congestion occurs when the communication channel or medium becomes overloaded, which results in degrading the network performance [2]. Network congestion is directly proportional to the vehicular density in a specific area. There is high mobility of nodes or vehicles, which is continuously changing the network topology and vehicular density. These dynamic topology changes create connectivity issues between V2V and V2I communication. In a dense network environment, significant numbers of nodes trying to send the packets cause congestion in the network. The delay and packet loss increase many folds when vehicular density increases resulting in



FIGURE 1. VANET environment.

network congestion [2], [3]. This consequently decreases the network throughput. Thus, congestion control becomes the point of interest to enhance the efficiency, reliability, and safety of the network.

Another important concern in VANETs is how to minimize communication delay achieving a low latency path. This is one of the crucial issues for the future deployment of VANETs. It is a point of concern that how to perform efficient computations with effective resource sharing that in turn decreases latency and enhances the quality of service (QoS) of connection [4], [5]. Since safety-critical messages require low latency so network congestion can have an impact on the latency of the transmitted packets. The critical messages face delays due to MAC contention and with the poor channel utilization, which is not suitable or desirable in VANETs [6]. Similarly, distribution coordination function (DCF) is a technique used to access medium in conventional 802.11 wireless local area networks, and analytical models are proposed to compute the throughput for DCF [7]. The contention-based carrier-sense multiple access with collision avoidance (CSMA/CA) scheme and a limited transmission range of DSRC make the dissemination of critical messages a challenge in VANETs [8]. The frequent delay in the dissemination of crucial messages can cause a severe threat to network performance and neighboring vehicles as well. Packet loss is another factor that affects the performance of the VANETs [9]. Packet loss occurs due to several different factors including packet transmission rate, transmission power, traffic flow, and communication distance.

In VANETs the key challenges are intermittent connectivity, high mobility, heterogeneous vehicle management, security, and support of network intelligence. Therefore, based on the above-mentioned concerns, designing optimized network protocols needs further exploration. This work is organized as Section II present important features of VANETs. Section III reviews different congestion controls strategies for VANETs. Section IV summarizes existing simulators used to simulate congestion scenarios. Section V concludes the work with the future perspectives.

#### **II. IMPORTANT FEATURES OF VANETS**

**IEEE**Access

In this section, we have discussed the main features of VANETs, which includes its characteristics, applications, and challenges. We have also discussed different performance metrics for congestion control at the end of this section.

## A. CHARACTERISTICS OF VANETS

VANETs itself is an application of mobile ad hoc networks (MANETs). Its highly mobile and dynamic nature makes it different from other mobile networks. The VANET characteristics include high mobility, dynamic topology, predictable mobility patterns, energy constraints, unbounded network size, wireless communication, etc.

- 1) **Dynamic topology** Vehicles move at high speed, and their direction and speed are continuously changing; this makes it difficult to predict vehicular position at the next instance. These abruptly changes the vehicular density in a particular region. With the increase in vehicular density, congestion control becomes a point of concern. With the varying speed, the geographical location of vehicular nodes also changes with time. This is the major difference between VANETs and fixed size network [10].
- 2) *Time criticality* One of the important features of information or messages in VANETs is its time criticality. The data and information should be delivered in time to take adequate actions. Let's say if medical emergency messages are delivered in time, it can save precious lives. That's why timely dissemination of safety-critical messages becomes crucial for network efficiency and reliability. The data or packets should be transmitted in a way that the congestion should be avoided increasing the throughput and decreasing the network latency [11], [12].
- 3) Mobility patterns Mobility patterns in VANETs are usually predictable as vehicles follow specific predefined pathways and roads. By analyzing the mobility patterns and information communicated among vehicular nodes, traffic jams, and other network scenarios can be predicted. This can help to predict the traffic condition in a region at a particular time. Similarly, by managing the mobility can help to effectively deliver the services to the vehicles [13], [14].
- 4) No energy constraints Fortunately VANETs do not have any power constraint like MANETs. The vehicle battery is sufficient to provide power for a variety of computations. It is viable to implement those schemes and solutions which are computationally intensive; however, the time-critical nature of information makes it challenging to come up with a reasonable solution that guarantees a tradeoff between efficiency and complexity. Secondly, the reliability of communication is an important concern to provide due assistance to drivers. The continuous acknowledgment of these messages can cause congestion [15].

- 5) Unbounded network size Unbounded network size means that VANET is not limited to a specific area, city or region but it can be implemented having variant sizes for a town, multiple cities, and even countries. It means you can develop a scalable version of VANET but high vehicular density along with higher mobility makes scalability a challenging issue. Bandwidth limitation makes it necessary to use the effective use of it. It is encouraged to design routing protocols that negate the principle of simple flooding preventing the broadcasting storm problem. This enhances the scalability to some extent [16].
- 6) Wireless communication and frequent information exchange – VANET is a wireless environment and Vehicles and nodes are continuously sharing the information. Vehicles receive periodic messages from other nodes as each vehicle wants to make other vehicles aware of the surroundings. Due to the wireless nature of VANETs, there are a lot of security concerns that need to consider for efficient communication. Another essential use-case and relatively new idea after the emergence of the internet of things (IoT) and VANETs is the smart cities or vehicles-to-everything (V2X) communication that will undoubtedly revolutionize the ways of interaction between machines. Secondly, handover also becomes the point of concern in these highly dynamic scenarios [17], [18].
- Better nodes protection Nodes in VANETs are better physically protected so this provides better physical security and is usually difficult to compromise. The on-board unit (OBU) provides sensors, resources, and read/write storage for data and user interface. This is responsible for communication between different OBUs and RSUs [19].

#### **B. VANETS APPLICATIONS**

VANET applications can be classified based on two essential requirements, named as efficiency and safety. Usually, VANET applications sense different types of data from the surrounding environment and take decisions by exchanging the relevant information with the other nodes. VANET application can be divided into safety applications, efficiency applications, comfort applications, interactive, entertainment, and urban sensing, etc. Multiple applications can be used by specifying their requirements. These applications are supported and used in a dynamic and multihop topology. Safety and non-safety applications are the most common ones while others are usually variations of it [20], [21].

 Safety applications – Safety application reduces the chances of an accident and is delay-sensitive. These applications provide early warnings to the vehicles to proactively preventing any event like an accident. More than 60 percent of the accidents can be avoided when a driver receives an early warning. This can reduce the accidents and congestion at the intersections by providing the drivers with the best route to the destination. Flow control at the intersection is challenging as many different flows are intersecting at that point. Safety messages are sent as cooperative awareness messages (CAM) which is a consequence of an emerging technology named cooperative intelligent transport system (C-ITS) [21], [22].

- 2) Efficiency applications These types of application make other vehicles aware of its location to enhance interaction between vehicles. It improves the vehicle's mobility-based on the information received from the neighboring vehicles. These types of applications enhance the perception of road traffic by predicting traffic. It uses different features like velocity, position, acceleration and other parameters of the neighboring vehicles to increase the cooperative awareness [23].
- 3) *Comfort or customization applications* –These applications make the driver's journey more comfortable and enjoyable. These applications usually include information about nearby restaurants, gas stations, weather information, etc. Suppose a driver's health condition becomes upset and wants to go to some hospital then the message can be sent to a vehicular cloud that replies with the location of the nearby health center. Similarly, in case of accidents, these types of applications can assist the driver. The vehicles can find the site of a bar or any friend meetup based on these applications. It can provide a crowdsourcing feature to enhance the interaction of applications for collective benefit [24].
- 4) Interactive applications These applications provide entertainment related services to passengers. Examples of this type of application include music, chats, web browsing, internet services, distributed games, etc. Let's say if a vehicle enters a new city, then different picnic spots and advertisements are downloaded automatically from the internet for passengers and drivers. Different multimedia files and entertainment services are attractive features of these applications. [25]–[27].
- 5) Urban sensing Sensing and communications can enhance the interactions and management in a traffic management system (TMS) [26]. The vehicles equipped with different kinds of sensors use their sensing capability to sense traffic conditions, video/audio surveillance, and environmental parameters. These applications sense the urban environment and their environmental conditions using OBU's sensors. This data of common interest is further shared among vehicles. The sensing of social activities play an important role in urban sensing [28], [29]. Similarly, application in VANETs can be divided into congestion road notification (CRN), cooperative collision warning (CCW), lane change assistance (LCA), post-crash notification, etc. [30].

# C. PERFORMANCE METRICS

Due to abrupt topology changes, the vehicular ad hoc network can exist for a short period. In this time, the data must be transmitted efficiently, and low latency paths must be established.

 emergansport control algorithm. Some of the commonly used metrics for performance measurements are discussed below [30].
 1) Average delay and average throughput – Average delay is the inter-packet delay (IDP) between successive packets. It is also called inter-reception time (IRT) which

ets. It is also called inter-reception time (IRT) which is reception time between two subsequent or consecutive sender recipient pair [32]. It is one of the most important parameters to determine the efficiency and effectiveness of the congestion control algorithm. The reason for this is the time criticality of the event-driven safety messages, as discussed earlier. The optimal flow of the network is based on the timely availability of these messages as actions are solely dependent on this critical information. Less inter packet or average delay is usually the indication of less congestion in the network. Similarly, vehicular density plays a vital role in latency and delays in transmitting packets [33]. It means more packets can reach the destination and fewer packets are wasted, dropped or delayed due to collisions or buffering. Throughput is the number of packets received which are sent from the source to destination. If throughput is low, it is an indication that the network is congested. Throughput and average delay are inversely proportional to each other if one is higher another one automatically becomes lower. Due to the unique nature of the vehicular network, the device discovery process time also needs to minimize, reducing the overall latency and delay [34].

Small changes in operational parameters lead to consider-

able effects on the efficiency of the network [31]. There are

numerous parameters to judge the efficiency of a congestion

- 2) Packet delivery ratio (PDR) is related to the number of packet send over the number of packets received [35]. PDR is an important parameter to determine the level of congestion in the network. The value of PDR lies between 0 to 1. A higher value of PDR means more packets can reach the destination. If the value of PDR is one, it means all sent packets have reached the destination. Low delay and high packet delivery are requirements for optimal VANET behavior. However, due to frequent changes in vehicular density and network topology, these parameters are drastically affected. By giving close attention to the time-varying behavior of VANETs can enhance the performance [36], [37].
- 3) Channel busy ratio (CBR) is an important parameter to enhance the packet delivery ratio which in turn will increase the packet delivery performance [38]. It is also referred to as channel busy time (CBT), which is the ratio of the time during which the channel is sensed busy and the total observation time. CBR is the measure of perceived congestion or channel load, which depends upon the number of vehicles in the transmission range and individual message generation rates. Usually, power and rate control strategies depend upon the channel busy ratio which is the feedback of the network against network traffic. These approaches try to maintain the busy

channel ration to an optimal level so that the network may not get congested [3].

4) *Parameters affecting performances matrices* – Different parameters affect the performance of the above parameters directly or indirectly. Similarly, vehicular speed, Vehicular density, area, and track length also affect the network congestion that is usually handled by varying the transmission range of vehicles [39]. The high speed makes the topology change more vibrant. Intersections are challenging to handle as compared to highway lanes.

# D. CHALLENGES FOR VANETS

In the previous section, we have discussed the application and characteristics of VANETs. In this section, we will see what the challenges associated with VANETs are. As previously discussed, the unique nature of VANETs has made it quite different from the traditional networks, and there are a lot of serious issues and challenges that need to be addressed in this type of network [40]. The associated challenges are:

- 1) *Delay constraints* A challenge in VANETs is to design a protocol that provides a low latency path with varying topology, speed, and connectivity. The network should provide adequate quality of service with fewer retransmissions, more extended connectivity, and little delay. By handling delay constraints effectively, the accidents can be reduced to a greater extent by enhancing the early decision making [41]. In an area where there are multiple vehicles transmitting beacons at the same time, it will create uneven delays causing network congestion. The speed of the vehicle is also one of the critical factors that can be adapted according to the situation to enhance network performance. [39].
- 2) Packets congestion control and prioritization Emergency packets have significant impact over the network performance. The network should not drop, delay or waste those packets. There is a significant amount of research done to find out how to give different priorities to these event-driven messages to enhance network performance. By giving priority to different packets based on their importance, the congestion and uneven delays in the network can be greatly reduced. It becomes the point of interest on how to give priority and develop priority queues based on the message types e.g. safety general, safety-critical, video etc [42].
- 3) Cross-layering and reliability VANETs use wireless medium to communicate. Due to wireless communication medium and high mobility, these connections and routes break frequently. It is challenging to design a protocol that effectively handles routing phenomena in these networks. In most of the reliable transmissions, a particular vehicle uses some scheduling discipline to transmit messages. Devising a concurrent transmission protocol is a challenge that can enhance network performance and reduces delays [43].

- 4) Geo addressing Some application requires the physical site of the vehicle to offer a different kind of services. Because of the high mobility and dynamic environment, the geo addressing is quite hard in VANETs. Most of the protocols support topology-based routing, but due to abrupt topology changes, these protocols are unsuitable for VANETs [44]. Position-based protocol is a solution to these problems but developing an efficient routing protocol in VANETs is a challenging task [45]. Often the protocols support unicast data forwarding in highly volatile ad hoc network.
- 5) Geo addressing Some application requires the physical site of the vehicle to offer a different kind of services. Because of the high mobility and dynamic environment, the geo addressing is quite hard in VANETs. Most of the protocols support topology-based routing, but due to abrupt topology changes, these protocols are unsuitable for VANETs [44]. Such protocols are a solution to these problems but developing an efficient routing protocol in VANETs is a challenging task [45]. Often the protocols support unicast data forwarding in highly volatile ad hoc networks.
- 6) Data trust and verification Trust and privacy are crucial concerns for the deployment for vehicular networks. VANETs communication broadly supports broadcast scenarios were verifying the authenticity of a particular message is very important [46]. Insecurity perspective usually, confidentiality is not an issue of prime importance as messages are broadcasted. But data verification is an essential aspect of data security. It is necessary to propose some solutions that ensure the integrity of data to predict better the future state of the network enhancing the decision making the process.

# E. MOTIVATION BEHIND THIS WORK

Congestion control is one of the most important concerns for VANETs. When traffic increases in a particular region, then the packet drop takes place, which is the result of the traffic congestion [47]. Every time a vehicle receives a message from the nearby cars, it updates its information and takes decisions based on that information. The collected data is crucial because the network state depends upon the availability and validity of these messages.

Every vehicle repeatedly sends CAMs that describe the current state of the network environment. These messages are sent using control channels. As CAMs messages are sent periodically so control channels are occupied by these messages for a significant amount of the time which can lead to congestion in the network [48]. These messages are crucial as they contain information regarding some accidents and congestion information at some intersection. Accidents and traffic congestion can be avoided by decreasing the delays of these crucial messages. If by any means network is congested, then these crucial messages will get wasted and remain unavailable to other vehicles at the right and desired instance. It can cause an uncertain future



FIGURE 2. Taxonomy of congestion control protocols.

state of the network with frequent accidents and uneven delays.

This work describes the approaches and strategies used to avoid network congestion so that efficient congestion control protocols may be designed by getting a broader familiarity with these schemes.

## **III. CONGESTION CONTROL STRATEGIES FOR VANETS**

In VANETs messages are broadcasted in the network. A vehicle creates a copy of a message and sends it to all other vehicles in a network. VANETs use three different approaches to address congestion in the network. [49], [50]. These approaches include proactive, reactive, and hybrid congestion control. Based on functional mechanisms and techniques, congestion control strategies using the approaches mentioned above in VANETs are mainly categorized into six types. The different types include rate-based, powerbased, hybrid strategies, CSMA/CA-based, prioritizing and clustering-based strategies [51], [52].

Power-based approaches try to prevent congestion by limiting the transmission power of nodes. When there are a large number of nodes sending and receiving packets at the same time, it will also increase the collision probability of packets causing the congestion and packet loss. This approach decreases the transmission power by a factor allowing less number of nodes transmitting or receiving the packets [3], [53]. Rate-based strategies shift the transmission or data rate based on the condition of the network. When the network is congested, and packet loss takes place, this type of protocol shifts the data rate of the medium to a lower rate to avoid collision of packets. That will eventually reduce the packet loss and congestion in the network [52], [54]. Hybrid approaches combine rate based and power-based methods and simultaneously change them to address congestion in the vehicular network. MAC-based strategies are default strategies to address congestion in VANETs. In CSMA/CA environment, a node detects the channel before sending a packet in the medium. If the medium is found free and no other node is transmitting the packet, it starts transmitting its packets. These types of protocols control channel access through adjustment of channel access parameters, e.g. contention window size [55]. In priority-based approaches, different priorities are assigned to messages based on their type [56]. Emergency or safety-critical messages are given high priority to get delivered to the destination as compared to general safety messages. In VANETs emergency or safety-critical information include accidental or congestion information that is crucial for the optimal performance of the network.

In this section, we will describe different strategies to overcome congestion in the network. The summarised taxonomy and related works are illustrated in 2.

### A. POWER-BASED STRATEGIES

Power-based strategies control congestion in the network by changing the transmission power based on the network behavior. If a vehicle sees the congestion in the network, it adjusts its transmission power accordingly to handle congestion [2], [57], [58].

1) Multi-metric tx-power control protocol (MPC) [59] - maintains an acceptable level of channel saturation and transmission power with the different coverage ranges. Control channel quality and application requirements are two factors to adapt the transmission power of each node according to its needs by effective use of control channels. It is important to know the channel quality for current and future transmissions to avoid channel saturation along with the collision. The current channel quality is calculated by measuring the CBT for the last 100ms. The protocol uses a beaconing load metric to calculate channel quality for future beacons. Denda et al. [60] explained the beaconing load is expected load determining how much load is there on the control channel for transmission power that covers the estimated vehicular density. Beacon load



FIGURE 3. Transmission power change based on vehicular densities.

for all vehicles is calculated by assuming a constant frequency and packet size. As far as the application requirement is concerned, transmission power is adapted based on the desired transmission range and priority for different types of messages (Figure 3). A higher priority level of 1 is assigned to event-driven messages, whereas a lower priority level of 0 is assigned to the safety general messages. The highest transmission power is used to send event-specific messages when there is no congestion in the network, whereas general purpose messages are sent with the lowest transmission power in case of congestion. In all other cases, transmission power is adapted to avoid congestion in the network. Drawbacks: On vehicular advertisement vehicular density is calculated. What is the threshold for congestion detection and how vehicles will receive beacons if there is already congestion in the network?

2) Speed-based adaptive power control (SBAPC) [60] reduces network congestion by adapting transmission power for basic safety messages (BSM) based on vehicle's speed. The protocol is based on the fact that increasing the speed of a vehicle decreases time to collision TTC) with neighboring vehicles. This means distant vehicles should have an awareness of a vehicle to receive BSMs. The protocol has initialization and power control phase. Maximum transmission power and cycle length are two metrics that are assigned with suitable values during the initialization phase. The value of 10mW is selected for maximum transmission power which corresponds to maximum distance for BSM to reach, i.e. 360m. Cycle length determines the number of BSM packets sent by a vehicle which is 7 in this case. The power control phase is invoked on each new cycle startup. Power factor is calculated based on maximum power level, cycle length and vehicle speed which eventually controls transmission power. After every BSM transmission, Vehicle's transmission power is increased by a factor based upon the power factor. The last BSM in the cycle is always sent with maximum transmission power irrespective of vehicle speed. The approach tries to expand the awareness circle for a vehicle increasing the transmission power. A quick increase in power

173202

level is observed in case of a higher speed. *Drawbacks:* A high vehicular density in a region will decrease the transmission power to a smaller value which will decrease the awareness of the surrounding vehicles.

- 3) Vehicle density-based power control (VDBPC) [61] - controls the transmission power on MAC layer based on the number of vehicles in a region. Initially in this maximum transmission range is selected, which is 1000m in this range and packets are set to transmit at the constant maximum rate of 10HZ. It means on average, 10 packets are transmitted for every second. Scheme counts the number of vehicles in a region and determines the state of vehicular density based on the count. If vehicle count is greater than 100, the state is considered dense. In moderately dense state vehicle count is between 50 to 100. On the other hand, if the vehicle count is less than 50, the scheme classifies it as a sparse topology. In dense environment transmission speed is shifted to a lower level because higher proximity will increase the collision rate causing congestion. In a moderate state, transmission power is shifted to a medium state such that it is just able to accommodate vehicles that are far apart. In the sparse state, transmission power is set to maximum as vehicles are far apart causing less congestion. Drawbacks: Congestion consequences for event-driven messages remained unaddressed. Vehicular count or density is randomly assumed.
- 4) Discussion The goal of the congestion control techniques is to make awareness of distant vehicles. Power-based approaches serve this purpose to increase the transmission range. It will make the remote vehicle better aware of the environment. The issue with the power-based approach is that in the case of higher transmission power scenarios, the hidden node, and channel fading phenomena become more prevalent. It reduces the channel sensing range as well. So power control approaches sacrifice the awareness of distant vehicles but can make the nearby vehicles better aware of its surroundings [62].

# B. RATE-BASED STRATEGIES

Rate-based strategies rely on changing the transmission rate or packet generation rate to control congestion in VANETs. Increasing the transmission rate has a significant effect on the performance of VANETs as it will increase the transmission rate of safety messages making vehicles more aware of their surrounding environment. There is also some cost attached to the increase in transmission rate leading to the congestion in the network. In the high-density scenario, there are large numbers of vehicles transmitting a large amount of data that increases the channel load and causes collision and channel saturation [1], [63], [64].

 Effective and efficient adaptive probabilistic data dissemination protocol (EEADP) [51] – incorporates distance from source node and number of road segments based on delay and probability to control congestion. This approach decreases the broadcast storm problem and increases the data delivery rate. The waiting period is calculated by using the distance variable, and the suitability of rebroadcasting is determined for each node based on its probability in the least waiting time. If there are large numbers of nodes on a road segment then increasing the number of road segments will eventually decrease vehicle density. Vehicle farthest from the source node is given the minimum wait time before transmitting. The broadcast problem can be resolved by giving different wait times to different segments with the last slot chosen as the unique forwarder node. If the number of vehicles in a region is increased, then the issue of rebroadcasting can be addressed by calculating the present and preceding redundancy which is the number of delivered messages to the total number of new messages. The slot with the lowest number of vehicles and redundancy ratio will be given the highest probability to rebroadcast based on the redundancy ratio. The approach is only suitable for safety messages with no periodic safety message exchange. Drawbacks: The proposed solution is cost-efficient. The author suggests that if vehicular density crosses a certain threshold, then the issue of network congestion can be addressed by introducing enough road segments. This will be quite inefficient in a particular scenario. There are numerous solutions like adjusting the transmission power, giving priority to event-driven safety messages that can be adopted to reduce the network congestion effectively.

2) Dynamic broadcast storm mitigation approach (DBSMA) [65] – senses any hazard efficiently and take immediate action accordingly. Perception time is 1/4 to 1/2, whereas reaction time is usually 1/4 to 3/4 of a second. A minimum safety distance of 200m is defined based on some safety rules at a speed of 120 km/h. Usually, 3s rule is observed if the vehicle is showing normal behavior in the network along with the minimum safety distance. The maximum communication range for the vehicle to vehicle communication is 1000m based on WAVE standard. A faulty vehicle parked on the roadside transmits messages to make other vehicles aware of the situation but transmits at the lower rate. A fast-moving vehicle moving at the speed of 120km/h transmits at a higher rate as compared to slow-moving vehicles. Tolerance range is 20% which is 800m in the specific scenario, and a parked vehicle or vehicles with speed less than 6km/h transmit at least two messages for the traveling distance of 800m. So fast-moving vehicles approaching the faulty or parked vehicle will slow down after receiving the messages, and it will start transmitting at a lower rate. This approach reduces congestion by varying the transmission rate according to the situation Drawbacks: The issue of unreachability of sensitive information can arise. The event-driven safety messages, i.e. accident information or traffic

- 3) Trajectory predicted mechanism (TPM) [66] is introduced to avoid congestion in V2I communication by reducing the beaconing rate. Congestion is avoided by reducing the messages that create vehicle awareness in the network. Each vehicle builds its future trajectory model based on its sensed information and previous predicted information. Rather than sending data at a constant rate, now, each car sends messages based on its predicted future trajectory. After some time, this model expires because of continuous topology change and the position becomes inaccurate. Now a new model is transmitted that provides updated predicted positions. TPM uses a linear model, and a new model is sent when a vehicle makes a turn, applies brakes or accelerates. A threshold for new model transmission is calculated based on the error position vector. Every vehicle has two position vectors: one received or gathered from the vehicle's GPS and the second is the predicted one. The difference between these two positions vector determines the error that is used as a threshold for a new updated model. TPM also forces vehicles to send a new model after a specific time even though estimated error remains within the threshold limit. Drawbacks: What if predicted trajectory is inaccurate due to the malicious behavior of some vehicles?
- 4) Discussion Data rate approaches can reduce the congestion to some extent by adopting the data rate according to the network behavior. When significant collisions are detected in the network, these schemes reduce the data rate that effectively controls the congestion. Data rate adjustments are usually tricky in the case of VANETs. The traditional network sends broadcast with least supported data rate, but the unbounded nature of VANETs makes it challenging to address the interference level with efficient usage of channels [67]. The hidden node terminal effect is amplified with the increase in vehicular density, and efficient channel utilization becomes quite challenging to allow a minimum level of interference. Similarly increasing the data rate decreases the transmission power for constant target loads and vice versa [68].

# C. HYBRID STRATEGIES

These approaches use the hybrid strategy to control power and rate at the same time. Rather than individually changing the power or rate, it uses an improved way to handle congestion. It adjusts the transmission rate based on the range of the transmission [69], [70].

1) *Environmental and context-aware combined power and rate distributed congestion control (ECPR)* [69] – improves cooperative awareness at the target distance by changing both transmission and data rate. The purpose of the congestion control is to improve awareness and make the other vehicles informed about the periodic or event-driven safety messages so that vehicles may prevent any hazardous action. Power is changed to address the awareness issue of neighboring vehicles and also reduces the channel load, whereas data rate control adjusts the data rate to make the efficient use of the resources by achieving the target load. If the data rate is increased, then the data rate is reduced and vice versa. Power is adapted based on the transmit power level of the sending vehicles which is piggybacked in the transmitted packets. Based on that, channel path loss for all vehicles is estimated by calculating the path loss exponent (PLE). To adjust the data rate, channel busy ratio and current beacon rate are used as an input to the linear message rate integrated control (LIMERIC), rate adaptation algorithm. The beacon data rate is adjusted to keep the channel busy ratio under the threshold. Drawbacks: As nodes are dynamic and mobile. Vehicles which piggyback the messages are not tracked so at the next instance topology will be changed, which can degrade the quality of awareness. Tracking error should be considered while implementing the congestion control [71].

- 2) Joint transmission power rate control (TPRC) [72] is a joint power/rate control that optimizes the performance of the reception for the vehicular safety communication by calculating the kth percentile of IRT at the target distance preventing the channel over-saturation. This is based on the fact that there is an optimal value of transmission power that independent of vehicular density, unlike the corresponding transmission rate. For this purpose, Spatio-temporal characteristics in terms of packet reception are analyzed for safety applications. The spatial aspect is related to this fact that sending vehicles should make sure that its beacons have listened at a target distance. The temporal element is related to IRT which is the maximum tolerable time for consecutive or sequential packets for particular sender and receiver. Performance is optimized by selecting the transmission power concerning the target distance, whereas channel load is used as a metric to choose the data rate reducing the unnecessary interference. The algorithm first checks if channel load is less than the maximum allowed load it will look for the maximum value of the transmission power for the maximum load in a lookup table and increases transmission power accordingly or increases the transmission rate is lower than the maximum tolerable limit. In the second scenario data rate or transmission power is decreased to optimize the reception performance. Drawbacks: Scheme only considers beacons and does not consider event-driven safety messages that are crucial for the safety of VANETs.
- 3) Combined power and rate control (CPRC) [70] addresses the congestion in the network in the single loop instead adjusting power and rate in two different phases. When a node comes across a dangerous

situation, which is turning left or right at the intersection, these nodes decrease their transmission power and allow other nodes to become aware of this dangerous situation. On the other hand, the transmission rate is increased required by some contextual factors like higher collision probability and speed such that it may remain below a predefined reliability-based threshold. In these schemes, two algorithms run such that after running the first algorithm second can intervene anytime providing a certain degree of flexibility. In the first step, the transmission rate is calculated based on vehicular density, and data rates are also determined for each transmits power level. Every node in a particular transmission range uses the same transmission rate to achieve fairness. In the second phase, each node calculates the inter-packet arrival time for its neighbors and also calculates its packet generation rate as well. So transmission rate is selected according to the transmission power such that overall traffic load is kept below a specific threshold. Drawbacks: Scheme uses the same data rate to achieve fairness. In practical scenarios fairness is not met with the same data rate. There are a variety of other factors like the vehicular speed that determine different data rates for different vehicles.

4) Discussion – These approaches mainly consider the power and data rate to address congestion in the network but ignore traffic dynamics and the status of vehicular movement, e.g. speed, etc. These approaches also do not take into account the position error computed by neighboring vehicles as the vehicular network requires the high accuracy of vehicles which can help to park vehicles at the specified place. Similarly, it also requires a better navigation facility to route the packets from source to destination.

# D. CSMA/CA-BASED STRATEGIES

CSMA/CA-based strategies are the default congestion control strategies in VANETs. These approaches try to address the congestion by changing the channel access parameters, e.g. congestion window (Figure 4). To provide services, each vehicle should be able to access the medium efficiently. MAC strategies describe how vehicles can efficiently access channel to reliably and effectively use safety and non-safety applications. Based on different types of media access MAC-based protocols can be divided into three types: contention-based, contention-free, and hybrid MAC protocols [73]–[76].

Binary exponential back-off (BEB) [77] – is a decrement exponential back-off strategy to find equilibrium between collisions and expired beacons while addressing the issue of ghost nodes. It reduces the contention window for the consecutive messages if the message is expired, improving the quality of communication on the control channel. The scheme discovers that RTS/CTS and BEB strategies are unsuitable for a broadcast environment like VANETs as CAM messages have limited lifetime and should be available at the right time to



FIGURE 4. CSMA/CA-based Strategies.

neighbors. If based on the contention window a vehicle has to wait for some time before sending any crucial message then that information becomes outdated after some time, it will be useless to the network and becomes hazardous. It develops the notion that for VANETs while considering CAM the achievement is not to send a large amount of data or increase the throughput, but the real objective is to make its neighboring vehicles more aware of its environment. Approach says that In case of optimal contention window in a crowded environment, the risk of losing some critical information always exists that cannot be tolerated. So genuine problem is how to achieve a tradeoff between collision and expired beacons? The tradeoff is achieved such that initially contention window (CW) is introduced with large value and then on each message expire window is reduced to half. By doing so, event-based crucial messages are given higher priorities with less collision and the issue of ghost nodes is tried to be controlled as well. Drawbacks: Reducing congestion window to half is not a good idea as the vehicle may be able to send packets more than the half without causing the collision.

2) Safety range-carrier sense multiple access (SR-CSMA) [78] – increases the reception probability of messages within the safety range by proposing a channel access technique. The safety range is larger than the transmission range of the vehicles. Congestion is avoided using carrier sense, contention window, and location of the transmitting vehicle. It uses the same concept of conventional CSMA. If no other node is transmitting and the channel is empty, it sends the beacon. If the channel is not empty, conventional CSMA starts an exponential backoff timer and waits for a specific time before transmitting the message again. SR-CSMA introduces an intermediary step before setting the waiting period. If the safety range of two transmitting nodes is nonempty and messages are crucial, then the channel is considered busy. If the channel is busy, this calculates the signal-to-interference ratio (SIR) to determine what level of interference of a node transmission in a channel

VOLUME 7, 2019

at the border of the safety range of the already transmitting station can still make the transmission possible increasing the reception probability. If SIR is greater than a threshold, then the channel is considered idle, and nodes can transmit its message. Drawbacks: If there are safety messages to send and the channel is busy, the random back-off timer makes it inefficient.

- 3) Predictive contention window (PCW) [79] determines the CW to mitigate broadcast collision in VANETs. It first classifies the vehicular information and then uses Bayesian prediction ranking (BPR) to find expected contention window for each attribute set using stochastic gradient and sigmoid function. After that hidden Markov model is used to find contention for the future by determining future vehicle state. Vehicular states are classified into attribute sets using classification strategy. The classification strategy uses a fuzzy nearness approach based on fuzzy mathematics to find a similarity degree between states of objective vehicles and attribute sets to determine window size. The window is selected based on multiple factors that represent vehicular density including neighboring vehicles, velocity and stop time, etc. BPR uses stochastic gradient descent (SGD) to find the expected value of each contention window corresponding to each attribute set. PCW Optimizes broadcast performance and reduces collision probability by implementing an estimation mechanism using a hidden Markov model (HMM). A vehicle sends traffic information on the control channel, and other vehicles adjust their window based on this information. HMM improves real-time ability by predicting vehicular state for future movement based on attribute sets and PCW adjusts window size supported via the MAC layer. Drawbacks: The predictive contention window predicts window size but does not address how it handles the safety messages.
- 4) Discussion These approaches try to sense medium before initiating their transmission. Because of the multi-access wireless medium, many factors can affect the transmission. If the vehicular density is increased in a region, then it can cause a delay in the transmission of safety messages because of the limited channel bandwidth. In the lower-density environment, CSMA/CAbased approaches perform better as they save the transmissions from wastage. It can make efficient use of bandwidth with fewer transmissions in a low-density environment.

## E. CLUSTER-BASED STRATEGIES

Clustering in VANETs is usually used to enhance the reliability and scalability of the network. In these techniques, vehicles are grouped into different clusters that in turn, make the vehicles to communicate in a distributed fashion (Figure 5). Each group is autonomously efficiently handling the responsibilities. These groups can better disseminate information and detect accidents or congestion in the network. The most important benefit is the scalability



FIGURE 5. Clustering-based approaches.

while going for cluster-based routing, where the network is converted from flat to hierarchical [80], [81]. Different clustering-based techniques include:

- 1) Unified framework of clustering (UFC) [82] uses mobility-based metrics to enhance the efficiency of the cluster. First of all, it separates stable neighbors that can build a connection with the cluster head (CH). A vehicle clears the criteria of a potential stable neighbor based on the similarity in the mobility pattern that includes the same direction and the speed difference smaller than a threshold. The overhead in clustering management is reduced through calculating the random backoff timer by each vehicle that makes a cluster head decision for that vehicle. A smaller backoff timer is selected by the vehicle that has a higher chance of being selected as a cluster head. Three parameters are considered For CH selection and computation of random backoff timer. These three parameters are average relative speed, average relative distance, and average link lifetime. The first vehicle sending the first cluster head broadcast announcement message is chosen as the first cluster head. Each vehicle communication with the cluster head maintains a cache of the list of cluster heads as it's back up, in case any connection to the cluster head fails then the vehicle can connect with the cluster head from the created list that is ordered based on their priorities. Backup cluster head with the longer link lifetime is given the highest priority. UFC is analyzed under the highly dynamic and stable traffic scenarios. Drawbacks: Random back-off timer-based schemes become inefficient for low latency networks like VANETs.
- Clustering-based probabilistic broadcasting (CPB) [83] – is clustering-based probabilistic broadcasting to efficiently disseminate safety messages for v2v communication in VANETs while reducing the collisions. A cluster head is selected based on the direction of the vehicle and its geographic location at a particular instance. Driving directions of vehicles make it easier to communicate effectively for the longer amount of time

between vehicles before discontinuity starts. After CH selection, each CH calculates its capacity of broadcast based on a locally generated counter to guarantee the transmission coverage and packet delivery ratio. For broadcasting the data, first, a probability assignment function is intelligently designed based on the vehicular density or the current condition of the network. A probabilistic forwarding algorithm is developed based on the probability assignment function that decides for data forwarding based on the calculated probability. If the packet receiver is the CH and is in the message dissemination range, it will broadcast the packet. On the other hand, if any other node has received the packets and CH does not reside in the transmission range of that node it will broadcast that packet probabilistically. In case CH is in transmission range, the vehicle drops the packet. It will, in turn, decrease the latency and the packet loss. Drawbacks: How to efficiently determine density threshold that can be used to forward packets on a probabilistic basis?

- 3) Vehicular multihop algorithm for stable clustering -Long term evolution (VMaSC-LTE) [84] – decreases end-to-end delay and increases packet delivery ratio by combining LTE with IEEE 802.11p based multi-hop clustering and minimizing the inter-cluster interference. Vehicles use average speed as a metric concerning other vehicles to select a CH. There are two interfaces defined: 802.11p operates between V2V communications, whereas LTE allows the CH to communicate with vehicles and infrastructure. The network creates a strong interference when clusters overlap in the space, which increases contention and collisions degrading network performance badly. This issue is resolved by keeping the size of the cluster and count to an optimal level. Each vehicle always tries to connect with the already present clusters and does not advertise itself a new CH unless necessary. Secondly, clusters in the transmission range of each other try to merge each other to form a single cluster with the numbers of vehicles smaller than the threshold. Initially, a vehicle connects with the cluster and tries to include it as cluster member, if cluster members are less than a maximum threshold. On the other side, vehicles send a multi-hop request to join the cluster if the cluster head is directly unreachable for that vehicle. If CH is indirectly unreachable as well than the vehicle can also advertise it as a CH. VMaSC-LTE proposes that neighboring cluster in the same direction can also be merged and slow-moving CH in previously clusters is selected as a new CH: Drawbacks: Does not explain the threshold or vehicular count for a cluster and its effects on the contention.
- A Cluster of CP-ABE Microservices for VANET [85]

   Deals with the computational delays about data encryption. It represents a clustering technique based on Kubernetes to efficiently encrypt the data with cipher policy attribute-based encryption (CP-ABE).



FIGURE 6. Priority and scheduling based strategy.

This, in turn, reduces the computational delays by distributing the encryption tasks to the vehicle cluster created with the help of Kubernetes. The encryption task is distributed based on the computational capabilities and resource information of the individual vehicles that Kubernetes collect to reduce the time of CP-ABE process. The encryption task is distributed in a V2V connection without using the infrastructure component, i.e. RSU. This decreases the computational cost of the operation reducing the delays. *Drawbacks:* If the number of clusters is increased, it also increases the inter-cluster interference. The scheme does not describe how to keep an optimal interference level when the number of clusters increases.

5) *Discussion* – clustering-based approaches enhanced the idea of access points in infrastructure-based networks where a node or CH just like an access point (AP) tries to share the network resources effectively with optimal scheduling of channel access [80]. In this way in which a large network is divided into smaller sub-network achieving better network management. The issues with clustering-based approaches are their cluster head management. All the information in the cluster depends on the expected behavior of the CH. If CH is failed then every function in the cluster will not perform as expected.

## F. PRIORITY-BASED STRATEGIES

Event-driven safety messages play a crucial role in vehicular networks. Crucial messages should be available to the destination at the right instant so that other vehicles and networks can adapt itself according to the network behavior (Figure 6). Some messages have higher priorities as they carry crucial information, e.g., accidental or congestion information. The network gives preference to these messages and loss, or late delivery of these packets can create havoc due to congestion. Below discussed are some protocols that try to handle congestion giving some messages priority over the other's [86], [87].

 Uni-priority safety messages dissemination using earliest deadline first (U-EDF) [88] – proposes a congestion control strategy to reliably deliver uni-priority or same priority event-driven messages on control channels. This protocol is divided into two parts. The first part is related to congestion detection and control, whereas the second part is the scheduling of safety messages. Congestion detection uses an event-driven and measurement-based detection method to measure traffic on control channels. In measurement-based detection, scheme packets are measured in a queue. If some packets are greater than a threshold that is five in this case, new incoming messages are dropped. Event-driven congestion detection is based on the fact that event-driven message is given higher priority as compared to ordinary packets. When these event-driven or crucial messages are detected, then all MAC transmission is free-zed, and these messages are provided with enough resources for their efficient dissemination in the network. The protocol uses the earliest deadline first (EDF) scheduling algorithm that is suited for real-time application. EDF handles messages based on their priority and deadlines that correspond to maximum latency. In VANETs each packet has attached deadline and priority, and packet with the smallest deadline is transmitted first by EDF. Drawbacks: This scheme only addresses uni-priority messages. Uni-priority messages are the messages that are generated from the traffic having the same priority level i.e emergency messages in this case. The scheme does describes to handle messages with different priorities.

2) Static and dynamic scheduling for congestion control (SDSCC) - decreases congestion in the network by prioritizing packets [84].It improves the IEEE 1609.2 multi-channel MAC on a prior basis by rescheduling the message queues before sending messages on control and service channels. In a static scheduling scheme, this approach makes a packet sending decision either to control or service channel based on the different static parameters and factors defined in the priority assignment unit. Messages of different types including beacons, safety low, and highest priority messages are sent to control channels, whereas lower priority messages are sent to service channels. In case the priority channel is congested traffic is diverted to the service channel. Dynamic scheduling further uses two methods. In the first method, initial priorities calculated by the priority transmission unit are rescheduled and packets in each queue are reordered when a new packet enters the queue. On the other hand, dynamic topology constraints make message scheduling an NP-hard problem. So, the second method uses a meta-heuristic algorithmic approach called tabu search algorithm that gives a near-optimal solution to schedule the control and service queues which reduces delay and jitter at the end of the day. Drawbacks: In normal circumstances the traffic on service channels is usually high as it carries the low priority or safety general messages. In case of congestion, the packets are directed to these channels. If service

channel is congested, where to send the event-driven safety messages.

3) Power efficient media access control (PE-MAC) [89] - modifies the original 801.11p protocol with PE-MAC protocol to set the congestion window and select a random back-off timer reducing delays for emergency messages. Data is divided into different types that include ambulance data, fire engine data, police car data, and normal data. Here ambulance data is the most crucial and sensitive one. When any data is transmitted, the data type is also appended with the data. The emergency data has a smaller back-off period as compared to other kinds of data. First, three types of data have normal distributions whereas normal data has a uniform distribution. Normal distribution picks the lower mean value that tends to have a smaller back-off timer with higher probabilities. Mean values are selected in order such that mean1>mean2>mean3, which ends up choosing the back-off timer in increasing order. If the average difference of packet sent to the MAC and passed to the link is less than the maximum delay for an emergency message transmission on the link than the contention window is divided into intervals. Otherwise means the value is reduced further and a smaller back-off timer is selected. Drawbacks: Retransmission is not a good option for low latency networks like VANETs. The delay and packet loss increases by the change in vehicular density which is the indication of network congestion.

# G. SDN-BASED STRATEGIES

- 1) SDN and fog-based intersection routing (SFIR) [90] Discusses an SDN and fog-based intersection routing (SFIR) approach that makes V2V communication possible in communication coverage holes. These communication coverage holes are the regions in the VANET environment where RSUs are not located, and vehicles take the responsibility to forward the data packets covering these holes. Intersection based routing tries to find the optimal path from the source to destination. Due to the lack of a global overview of the system, some times optimal routing paths are not generated. This issue is resolved here by using the software defined networking (SDN). Vehicles density, street length and Euclid distance among vehicular nodes are calculated by the fog nodes in coverage zones and coverage holes aswell. The gathered information is used by the SDN controller to find the optimal and appropriate transmission paths. Data is forwarded on the V2V channel using the computed paths. Drawbacks: There can be numerous available metrics to consider in these kinds of scenarios. How many parameters or features can give an optimal solution is still an open problem.
- SDN-Enabled Social-Aware Clustering algorithm (SESAC) [91]– Proposes clustering techniques based on software-defined networking (SDN) using 5G VANETs to decrease packet loss and congestion. Due

173208

to the flexibility and programmability of the SDN, it enhances the stability of the cluster in the highly dynamic VANET environment. The stability of the cluster is enhanced by exploiting the future route predictions or social patterns in VANET. Social patterns are defined as the corresponding sojourn period on each road segment along with the number of those road segments in which a vehicle is going to travel. Sojourn time probability distribution and state transition probability are passed as an input to the discrete time-homogeneous semi-Markov model which gives the social pattern of each vehicle as an output. These predicted outputs are used to develop clusters such that all vehicles in a cluster share the same route. Cluster heads(CH) are chosen based on different metrics which include relative speed, vehicle distance and other attributes of the vehicle. Drawbacks: The optimal behavior of the cluster depends upon many other factors as well. As the number of cluster increases, it also increases the inter-cluster interference. So what should be an optimal number for the cluster is an important aspect to deal with.

A summary table with all the congestion control approaches and their common attributes is shown in Table 1.

## **IV. VEHICULAR NETWORK SIMULATION**

Simulations provide a cost-effective solution to model or simulate the performance of objects or activities in a real-time environment. Sometimes due to nature and design constraints, practical experiments cannot be performed, so it is better to simulate the environment that imitates the physical nature of your environment. There are a variety of simulation tools available to simulate vehicular networks, including open source and commercially available simulation tools [70], [92]. Simulation in VANETs is mainly performed with two different types of simulators, namely mobility generator or traffic simulators and network simulators. Network simulators are used to design and simulate protocols for VANETs whereas traffic simulators are usually used to manage and simulate traffic or road behavior for these protocols. Mobility generator specifically addresses the mobility aspect of the vehicles which depends upon the track length, vehicular speed, vehicular density and number of lanes, etc. These simulators can be used individually but to model and simulate a real VANET environment both of these simulation tools are used collaboratively. VANETs use different types of VANETs simulators to make mobility generators and network simulators work in a collaboration, which we will discuss shortly. In the following section, we will discuss these three different types of simulation tools, their characteristics along with advantages and disadvantages.

#### A. TRAFFIC SIMULATORS/MOBILITY GENERATORS

As discussed earlier, that mobility generators primarily focus on the mobility aspect. Several simulators perform traffic engineering of vehicles in large geographical areas and cities. There are numerous traffic simulators available and

#### TABLE 1. Summary table of congestion strategies with relevant attributes.

	Simulator	Dataset	Density	Delay	Throughput	Loss	Coverage	Nodes/ devices	Internode commun.	Commun. protocol
Power-based strategies	s									
Shah et al. (2016)	OMNeT++/ Veins	U/L	~	~	$\checkmark$	×	750m	×	~	DSRC
Joseph et al. (2018)	OMNeT++/ Veins/SUMO	D/U/M	~	~	$\checkmark$	×	1000m	×	~	DSRC/Wave
Akinlade (2018)	OMNeT++/ Veins/SUMO	D/U/M	~	~	$\checkmark$	~	1000m	×	~	DSRC/Wave
Rate-based strategies										
Sospeter et al. (2018)	NS2/SUMO	U/ML	$\checkmark$	$\checkmark$	$\checkmark$	×	700m	×	$\checkmark$	IEE802.11p
Feukeu et al. (2018)	MATLAB	D/U/ML	~	~	$\checkmark$	×	1000m	×	$\checkmark$	IEEE802.11p/IEEE1609
Boquet et al. (2017)	NS3/SUMO	D/U/ML	~	~	~	×	20 mW	~	~	IEEE802.11p
CSMA/CA-based stra	tegies									
Stanica et al. (2011)	JiST/ SWANS/STRAW	D/HW	$\checkmark$	~	$\checkmark$	~	200m	×	$\checkmark$	IEEE802.11p
Stanica et al. (2012)	NA	U	~	~	$\checkmark$	×	100m	×	$\checkmark$	IEEE802.11p
Y.LU et al. (2016)	NA	D/U	$\checkmark$	$\checkmark$	$\checkmark$	×	NA	×	~	IEEE802.11p
Clustering-based strat	egies									
Ren et al. (2018)	NS2/SUMO	HW/M	×	~	$\checkmark$	×	300m	×	~	IEEE802.11p
Liu et al. (2018)	NS2/ VanetMobiSim	HW/H	~	~	$\checkmark$	×	250m	×	$\checkmark$	IEEE802.11DCF
Ucar et al. (2016)	NS3/SUMO	U/H	$\checkmark$	$\checkmark$	$\checkmark$	×	200m	×	$\checkmark$	IEEE802.11p-LTE
Taha et al.(2019)	NA	NA	×	$\checkmark$	×	×	NA	~	$\checkmark$	NA
Priority-based strateg	ies									
Darus et al. (2013)	OMNeT++/ SWANS/STRAW	Н	~	~	$\checkmark$	×	400m	×	~	IEEE802.11p
Taherkhani et al. (2016)	NS2/SUMO/ MOVE	U/H	~	~	$\checkmark$	×	10 packet/s	×	~	IEEE802.11p
Nellore et al. (2016)	NS2/MATLAB	NA	$\checkmark$	$\checkmark$	$\checkmark$	×	NA	×	$\checkmark$	IEEE802.11p
SDN-based strategies										
Noorani et al. (2018)	NS2/SUMO	U/M	×	~	$\checkmark$	×	200m	×	$\checkmark$	IEEE802.11p
Qi et al. (2018)	NA	U/M	~	~	×	×	350m	×	~	IEEE802.11p
Hybrid strategies										
Aygun et al. (2016)	NS2/GEMv2 V2V/SUMO	U	~	~	$\checkmark$	×	500m	×	$\checkmark$	DCC/IEEE802.11p
Tielert et al. (2013)	NS3	U	~	~	$\checkmark$	×	600m	×	~	NA
Baldessari et al. (2019)	NS2	HW/H	~	~	~	×	500m	~	~	IEEE802.11MAC
D: Dense, U: Urban, H	W: Highway, ML: Mu	ltilane, L: Lo	w Speed, M:	Medium s	peed, H: High s	peed, NA	: Not available			

most commonly used are SUMO, MOVE, STRAW, vanet-MobiSim, PARAMICS, VISSIM, and MoNoTrac.

- Simulation of urban mobility (SUMO) [93] is open source and one of the widely used traffic simulator which can handle significant traffic. SUMO supports vehicle collision avoidance, multiple vehicle categories, traffic portability, and speed control, etc. It can import as well as edit different mobility maps. SUMO uses GUI to route traffic through single and multiple vehicles. It can be used with NS2 NS3 and OMNeT++.
- Mobility model generator for vehicular networks (MOVE) [94] – is GUI-based build on the top and an extension of SUMO. It saves one's time to write

simulation scripts to simulate the environment. MOVE edits maps, and it can be used with other simulators like NS2 and global mobile information system simulator (GloMoSim).

- 3) Street random waypoint (STRAW) [95] is only used with SWANS. Modifications are usually performed to use it with NS2 or some other simulators. It can parse the TIGER map, but its management is complex, due to this reason it is not widely used.
- vanet-MobiSim [96] is Java-based mobility generator which is used with NS2, GloMoSim, and QualNet. It supports different intelligent mobility models like lane changing and intersection management. It can also

use pre-compiled intelligent mobility models as well. This simulator simulates a realistic mobility model, and besides generating those models, it can also import different US mobility models.

- 5) *PARAMICS* [97] is commercially available and can be used by adding plugins for macroscopic or microscopic models. It provides enhanced features like 3D visualization and large traffic generation.
- 6) *VISSIM*<sup>1</sup> simulates different types of traffic scenarios like vehicles, pedestrians, public transport etc. Inherently it addresses microscopic traffic flow, but it can support multi traffic flows which include microscopic and macroscopic traffic flow levels. That's why it is also called a hybrid simulator. Its output can be exported to some 3D graphics platforms, e.g. AutoCAD.
- 7) *Mobile node trace generator (MoNoTrac)* [98] is a Java-based traffic simulator which can create a simulation environment based on real geographic data. The user can define different simulation parameters like vehicle position and simulation time on its own. It can also be used with NS2, NS3, and OMNeT++ with extensible markup language (XML) formats.

# **B. NETWORK SIMULATORS**

Network Simulators are used to analyze the performance of different protocols under different scenarios in a controlled environment. We are specifically interested in network simulators that support DSRC 802.11 standard for vehicular communication. Most of the popular commonly used vehicular network simulators are NS2, NS3, OMNeT++, Mininet, and OPNET.

- Network simulator 2 (NS2) [99] is an open-source easy to implement discrete-event network simulator. NS2 uses object-oriented tool command language (OTCL) for simulation modeling with a simulation kernel written in C++. The initial version of NS2 has no support for IEEE802.11p but in later versions, e.g. NS2.33 the support was included. NS2 is extended to include radio propagation model and node mobility. It uses an event log that is supported by an event scheduler. Pros: It is open-source, and its setup is easy. Cons: NS2 scalability aspect is poor; it does not support scalable architecture. Initial versions of NS2 do not support VANETs default IEEE802.11p wireless standard but later version support this, does not support parallel processing as well.
- Network simulator 3 (NS3) [100] is an enhancement on NS2 and still under development. It is purely written in C++ which removes the OTCL for simulation modeling from NS2. It has significantly reduced the code length to a smaller value. It also enables the python script support. Pros: It supports multiple programming languages like C++ and python, including the default IEEE802.11p standard. NS3 also

supports parallel processing and distributed simulations. Its throughput is high. Cons: NS3 has a higher network delay with different routing protocols like AODV and DSDV because many packets cannot be delivered to the destination at once.

- 3) Objective modular network test-bed in C++(OMNeT++) [101] – Is C++ based, free and open-source discrete-time event simulator which can be used for the simulation of different communication networks using a GUI. The simulation system in OMNeT++ is developed by integrating different modules. There are two major frameworks currently used in OMNeT++ including INIT and mobility framework [104]. The mobility framework describes different layers of the internet protocol stack, whereas, on the other hand, the INIT framework describes different mobile and wireless protocols. INIT supports different MAC models like IEEE802.11a, but it does not support the IEEE802.11p model. OMNeT++ output, which is a text file, can also be processed with MATLAB and other simulation tools. Its open-source version is only available for nonprofit, and academics use, other versions are available for commercial use. Pros: Computation time or simulation runtime is faster. OMNeT++ has quite less network delay, and it has improved performance as per the network delay is concerned. As a simulation tool, it is memory efficient. Cons: Does not support scalability and throughput decreases as the number of nodes increases.
- 4) Mininet<sup>2</sup> [102] is an open-source network emulator that can emulate an environment with the help of an extensible python APIs. It makes each networking switch and node to work as a real-time machine by enabling virtualization. Mininet can also simulate a wireless network by adding Mininet wi-fi add-ons. It supports the IEEE802.11p protocol and can simulate wireless networks with SDN. Pros: It is an emulator that offers virtualization and effective resource sharing. Cons: Scalability of Mininnet lies in the middle. It lacks the performance fidelity as due to the emulated environment data forwarding rate of open flow switches is unpredictable. Due to this reason, it only helps to analyze the network behavior, not network performance [105].
- 5) *Optimized network engineering tools (OPNET)* [103] – is discrete-time event simulator that uses GUI to configure parameters for different layers and defines topologies. GUI inputs are mapped to simulate the real systems using an objected-oriented paradigm. It can make use of external libraries with its open interface and also supports custom packet formats. Pros: Supports custom packet formats and deal with external libraries. OPNET supports different programming languages like C++ and Java. Its GUI support is excellent. It also supports

<sup>&</sup>lt;sup>1</sup>http://vision-traffic.ptvgroup.com/en-us/products/ptv-vissim/

#### TABLE 2. Comparison of simulation software.

	Written in	Language support	Multi-language support	802.11p support	Scalability	Network delay	Availability	Learning curve	Virtualization support	Parallel support
NS2 [103]	C++	OTCL/C++	$\checkmark$	$\checkmark$	Low	High	~	Easy	×	×
NS3 [104]	Python/C++	C++	$\checkmark$	$\checkmark$	Limited	Low	$\checkmark$	Hard	$\checkmark$	$\checkmark$
OMNeT++ [105]	C++	C++	×	$\checkmark$	Medium	Medium	$\checkmark$	Easy	×	$\checkmark$
Mininet [106]	Python	Python	×	$\checkmark$	Medium	NA	$\checkmark$	Hard	$\checkmark$	$\checkmark$
OPNET [107]	Č++	C++/OTCL	$\checkmark$	$\checkmark$	large	Low	×	Easy	×	$\checkmark$
JiST/SWANS [106]	Java	Java	×	×	High	Low	$\checkmark$	Hard	$\checkmark$	$\checkmark$

#### TABLE 3. Evaluation measures used.

Author (Year)	Delay	Throughput	Loss	Contention	Energy	Clustering overhead	Coverage
Shah et al. (2016)	$\checkmark$	$\checkmark$	×	×	×	×	×
Joseph et al. (2018)	$\checkmark$	$\checkmark$	×	$\checkmark$	×	×	×
Akinlade (2019)	$\checkmark$	$\checkmark$	~	$\checkmark$	×	×	×
Sospeter et al. (2018)	$\checkmark$	~	×	~	×	×	×
Feukeu et al. (2018)	$\checkmark$	$\checkmark$	×	×	×	×	×
Boquet et al. (2017)	$\checkmark$	$\checkmark$	×	$\checkmark$	×	×	×
Stanica et al. (2011)	$\checkmark$	$\checkmark$	$\checkmark$	~	×	×	×
Stanica et al. (2012)	$\checkmark$	$\checkmark$	×	×	×	×	×
Y.LU et al. (2016)	$\checkmark$	$\checkmark$	×	×	×	×	×
Ren et al. (2018)	$\checkmark$	$\checkmark$	×	×	×	×	×
Liu et al. (2018)	$\checkmark$	$\checkmark$	×	×	×	×	×
Ucar et al. (2016)	$\checkmark$	~	×	×	×	~	×
M.B.Taha et al.(2019)	$\checkmark$	×	×	×	$\checkmark$	×	×
Darus et al. (2013)	$\checkmark$	$\checkmark$	×	~	×	×	$\checkmark$
Taherkhani et al. (2016)	$\checkmark$	$\checkmark$	$\checkmark$	×	×	×	×
Nellore et al. (2013)	$\checkmark$	$\checkmark$	×	×	$\checkmark$	×	$\checkmark$
N.Noorani et al. (2018)	$\checkmark$	$\checkmark$	×	×	×	×	$\checkmark$
W.Qi et al. (2018)	$\checkmark$	×	×	×	×	×	$\checkmark$
Aygun et al. (2016)	$\checkmark$	~	×	$\checkmark$	×	×	×
Tielert et al. (2013)	$\checkmark$	$\checkmark$	×	×	×	×	×
Baldessari et al. (2019)	~	$\checkmark$	×	$\checkmark$	×	×	×

communication with other protocols. Cons: It is complex and takes time to master. OPNET is commercially available, so it is expensive.

6) *Java in simulation time (JiST)* [102] – Is built on the top of Java virtual machine and can simulate larger scale wireless sensor networks with less memory and computational resources as compared to NS2. It is a discrete-time event simulator that uses SWANS wireless network simulation engine built on the JiSt platform to simulate network. Pros: Benefit scalability. Cons: Does not have 802.11p support which is the default standard.

## C. VANET SIMULATORS AND FRAMEWORKS

In this section, we will discuss different frameworks that integrate mobility generators and simulators to simulate an environment. So it forms an integrated environment that resolves the issues of interdependencies. Famous VANET frameworks are Veins, VSIMRTI, VENTOS, etc.

VOLUME 7, 2019

- Veins [106] is a popular integrator framework that integrates SUMO mobility generator with INIT framework in OMNeT++ simulator. Veins allow network simulator to communicate with mobility generator. Mobility generator slows down or controls the flow of traffic based on warning messages generated by the network simulator to control the network congestion. Messages are transmitted on TCP. There is also an integrated application that controls the emission of carbon dioxide.
- 2) V2X simulation runtime infrastructure (VSIM-RTI) [64] – supports a layered architecture and can be used to simulate latest ITS application using VSIM-RTI AppNT module. It has a flexible framework and easy integration that's why it can get integrated with a variety of network simulators like NS3, JiST/SWANS, and OMNeT++. VSIMRTI also supports a couple of mobility generators, e.g. SUMO and VISSIM, etc.
- 3) Vehicular network open simulator (VENTOS) [107] – is cooperative platoon management simulator that

uses OMNeT++ and SUMO as network simulator and mobility generator respectively. It uses the DSRC standard to enable V2I communication. Bidirectional communication is made possible by using the SNMP protocol. Other supported features include dynamic routing of traffic and the adversary models for security attacks.

### D. SUMMARY AND PERFORMANCE ANALYSIS

This section mainly discusses the performance aspects of discussed congestion control strategies. As previously discussed congestion control is a crucial concern in VANETs. There are varieties of parameters that can become the point of interest while addressing congestion in the vehicular network. The two most important parameters are latency and throughput. Latency discusses how much time a message takes moving from source to destination. Whereas throughput addresses how much data is transmitted from source to destination in a unit time. If latency is reduced and throughput is increased, then this is the desired network behavior. If resources are optimally distributed and congestion control is addressed efficiently, it can significantly enhance the performance of VANETs. In-fact highly dynamic nature of VANETs makes it a very challenging task to set these two parameters at an optimal level. We have considered six prominent congestion control strategies and discussed different kinds of protocols using these strategies.

Proactive congestion control or open-loop approaches are best suited for low latency networks. These types of routing approaches take action priorly before the occurrence of congestion in the network taking preventive measures against congestion. Reactive congestion control or closed-loop methods try to detect congestion after it has occurred. These approaches introduce delays and take action after the network has become congested. So on the one hand, delay and throughput are drastically affected, and secondly, high dynamic topology quickly makes network condition, even worse, at every new instance degrading the network performance severely. Proactive approaches, on the other side, try to address congestion before it's happening and give the network enough time not to go into a deadlock state.

Out of six discussed protocols, power-based and rate-based approaches take feedback from the network to adjust their transmission power or data rate accordingly. It means if congestion increases more than a threshold, preventive measures are taken to overcome congestion by adjusting transmission power and data rate. Hybrid approaches deal with the transmission power and rate at the same time to address congestion phenomena. Albeit 802.11P uses CSMA/CA-based approach but increase in the vehicular density degrades the VANETs performance badly. It just behaves like a random transmission technique rather than properly sending the data at higher densities. On the other hand, priority-based and clustering-based congestion control protocols show better performance compared to previously discussed approaches. These techniques tend to have increased throughput and decreased latency, providing better congestion control. The clustering-based method uses a distributed approach to make the network into smaller subgroup which causes a better strategy to handle packet collisions and congestion in the network. Priority-based techniques give individual packets higher priority as compared to others increasing their chances to reach destinations. A summary of the evaluation measures used by the different approaches is shown in Table 3.

## V. CONCLUSION AND FUTURE WORK

This paper is a survey of different congestion avoidance protocols and techniques for VANETs. We discussed different congestion control strategies and analyzed different important network performance parameters, including throughput and latency. We also discussed prominent simulators support used to simulate network environment along with their pros and cons. We have seen that proactive approaches are best suited for congestion control in VANETs. Due to their proactive congestion handling nature, priority-based and clustering-based approaches show enhanced performance as compared to other congestion avoidance approaches if hybrid strategies are Incorporated along with priority-based strategy, it can even produce better results.

As per the future work is concerned, cloud-based approaches with artificial intelligence techniques can be a promising research area for efficiently handling network congestion. Fog computing is an active research area that is a suitable choice for low latency networks like VANETs. Rather than sending the data for processing to cloud it can be processed by sending to fog servers that are placed near vehicles. Using AI with fog computing will have a couple of benefits.

First AI-based techniques can predict physically congested areas in the future so that based on the prior future predictions; it can effectively divert the traffic to other directions or tracks. Less vehicular density in a region will decrease the congestion in the network. Secondly, most of the discussed approaches have emphasized more on congestion control by controlling network traffic. In-fact low latency for crucial or event-driven safety messages is the core issue as compared to merely managing the traffic flow. AI along with fog computing can separate those event-driven or crucial messages from the ordinary traffic that can be processed in a better way for their efficient dissemination in the network. An interesting concern is if any malicious entity in the network individually or a group tries to consume network resources to create network congestion, e.g. DOS attack, AI can efficiently handle all these concerns by analyzing the vehicle or node behavior.

#### REFERENCES

- M. Torrent-Moreno, J. Mittag, P. Santi, and H. Hartenstein, "Vehicleto-vehicle communication: Fair transmit power control for safety-critical information," *IEEE Trans. Veh. Technol.*, vol. 58, no. 7, pp. 3684–3703, Sep. 2009.
- [2] N. Taherkhani and S. Pierre, "Centralized and localized data congestion control strategy for vehicular ad hoc networks using a machine learning clustering algorithm," *IEEE Trans. Intell. Transp. Syst.*, vol. 17, no. 11, pp. 3275–3285, Nov. 2016.

- [3] Y. P. Fallah, N. Nasiriani, and H. Krishnan, "Stable and fair power control in vehicle safety networks," *IEEE Trans. Veh. Technol.*, vol. 65, no. 3, pp. 1662–1675, Mar. 2016.
- [4] V. Balasubramanian, S. Otoum, M. Aloqaily, I. Al Ridhawi, and Y. Jararweh, "Low-latency vehicular edge: A vehicular infrastructure model for 5G," *Simul. Model. Pract. Theory*, vol. 98, Jan. 2020, Art. no. 101968.
- [5] M. Aloqaily, V. Balasubramanian, F. Zaman, I. Al Ridhawi, and Y. Jararweh, "Congestion mitigation in densely crowded environments for augmenting QoS in vehicular clouds," in *Proc. 8th ACM Symp. Design Anal. Intell. Veh. Netw. Appl.*, Oct. 2018, pp. 49–56.
- [6] U. A. Khan and S. S. Lee, "Multi-layer problems and solutions in VANETs: A review," *Electronics*, vol. 8, no. 2, p. 204, 2019.
- [7] G. Bianchi, "Performance analysis of the IEEE 802.11 distributed coordination function," *IEEE J. Sel. Areas Commun.*, vol. 18, no. 3, pp. 535–547, Mar. 2000.
- [8] B. Liu, D. Jia, J. Wang, K. Lu, and L. Wu, "Cloud-assisted safety message dissemination in VANET—Cellular heterogeneous wireless network," *IEEE Syst. J.*, vol. 11, no. 1, pp. 128–139, Mar. 2017.
- [9] Y. Liu, K. Shi, G. Xu, S. Lin, and S. Li, "Analysis of packet loss characteristics in VANETs," in *Proc. 8th Int. Conf. Electron. Inf. Emergency Commun. (ICEIEC)*, Jun. 2018, pp. 219–222.
- [10] H. Zhang and J. Li, "Dynamical topology analysis of VANET based on complex networks theory," *Cybern. Inf. Technol.*, vol. 14, pp. 172–186, Dec. 2014.
- [11] S. S. Shah, A. W. Malik, A. U. Rahman, S. Iqbal, and S. U. Khan, "Time barrier-based emergency message dissemination in vehicular adhoc networks," *IEEE Access*, vol. 7, pp. 16494–16503, 2019.
- [12] Y. Zhuang, J. Pan, Y. Luo, and L. Cai, "Time and location-critical emergency message dissemination for vehicular ad-hoc networks," *IEEE J. Sel. Areas Commun*, vol. 29, no. 1, pp. 187–196, Jan. 2011.
- [13] S. A. Vaqar and O. Basir, "Traffic pattern detection in a partially deployed vehicular Ad Hoc network of vehicles," *IEEE Wireless Commun.*, vol. 16, no. 6, pp. 40–46, Dec. 2009.
- [14] W. Ma, Y. Fang, and P. Lin, "Mobility management strategy based on user mobility patterns in wireless networks," *IEEE Trans. Veh. Technol.*, vol. 56, no. 1, pp. 322–330, Jan. 2007.
- [15] Z. Shafiq, S. A. Mahmud, G. M. Khan, H. Zafar, and H. S. Al-Raweshidy, "A complete transmission acknowledgement scheme for VANETs," in *Proc. 13th Int. Conf. ITS Telecommun. (ITST)*, Nov. 2013, pp. 238–243.
- [16] A. Ahamed and H. Vakilzadian, "Issues and challenges in VANET routing protocols," in *Proc. IEEE Int. Conf. Electro/Inf. Technol. (EIT)*, May 2018, pp. 723–728.
- [17] A. Gharsallah, N. Omheni, K. Ghanmi, F. Zarai, and M. Neji, "A seamless mobility mechanism for V2V communications," in *Proc. IEEE/ACS 14th Int. Conf. Comput. Syst. Appl. (AICCSA)*, Oct./Nov. 2017, pp. 1063–1069.
- [18] R. Zagrouba, T. Hayouni, and F. Kamoun, "Handover optimization within vehicular networks," in *Proc. World Congr. Comput. Appl. Inf. Syst.*, Jan. 2014, pp. 1–5.
- [19] R. Barskar and M. Chawla, "Vehicular ad hoc networks and its applications in diversified fields," *Int. J. Comput. Appl.*, vol. 123, pp. 7–11, Aug. 2015.
- [20] F. Cunha, L. Villas, A. Boukerche, G. Maia, A. Viana, R. A. Mini, and A. A. Loureiro, "Data communication in VANETs: Protocols, applications and challenges," *Ad Hoc Netw.*, vol. 44, pp. 90–103, Jul. 2016.
- [21] Y. Toor, P. Muhlethaler, and A. Laouiti, "Vehicle ad hoc networks: Applications and related technical issues," *IEEE Commun. Surveys Tuts.*, vol. 10, no. 3, pp. 74–88, 3rd Quart., 2008.
- [22] M. A. Javed and E. B. Hamida, "On the interrelation of security, QoS, and safety in cooperative ITS," *IEEE Trans. Intell. Transp. Syst.*, vol. 18, no. 7, pp. 1943–1957, Jul. 2017.
- [23] P. Lytrivis, G. Thomaidis, M. Tsogas, and A. Amditis, "An advanced cooperative path prediction algorithm for safety applications in vehicular networks," *IEEE Trans. Intell. Transp. Syst.*, vol. 12, no. 3, pp. 669–679, Sep. 2011.
- [24] A. M. Vegni and V. Loscrí, "A survey on vehicular social networks," *IEEE Commun. Surveys Tuts.*, vol. 17, no. 4, pp. 2397–2419, 4th Quart., 2015.
- [25] N. Hegde and S. S. Manvi, "Emerging vehicular cloud applications," in Proc. Int. Conf. Comput. Commun. Inform. (ICCCI), Jan. 2017, pp. 1–6.
- [26] S. Djahel, R. Doolan, G.-M. Muntean, and J. Murphy, "A communications-oriented perspective on traffic management systems for smart cities: Challenges and innovative approaches," *IEEE Commun. Surveys Tuts.*, vol. 17, no. 1, pp. 125–151, Mar. 2015.

- [27] H. T. Cheng, H. Shan, and W. Zhuang, "Infotainment and road safety service support in vehicular networking: From a communication perspective," *Mech. Syst. Signal Process.*, vol. 25, no. 6, pp. 2020–2038, Aug. 2011.
- [28] U. Lee, E. Magistretti, M. Gerla, P. Bellavista, and A. Corradi, "Dissemination and harvesting of urban data using vehicular sensing platforms," *IEEE Trans. Veh. Technol.*, vol. 58, no. 2, pp. 882–901, Feb. 2009.
- [29] D. Cuff, M. Hansen, and J. Kang, "Urban sensing: Out of the woods," Commun. ACM, vol. 51, no. 3, pp. 24–33, Mar. 2008.
- [30] A. M. Vegni, M. Biagi, and R. Cusani, "Smart vehicles, technologies and main applications in vehicular ad hoc networks," in *Vehicular Technologies: Deployment and Applications*. L. G. Giordano and L. Reggiani, Eds., London U.K.: IntechOpen, 2013.
- [31] T. Hewer, M. Nekovee, and P. V. Coveney, "Universality of performance parameters in vehicular ad hoc networks," *IEEE Commun. Lett.*, vol. 15, no. 9, pp. 947–949, Sep. 2011.
- [32] H. Chang, Y. E. Song, H. Kim, and H. Jung, "Distributed transmission power control for communication congestion control and awareness enhancement in VANETs," *PLoS ONE*, vol. 13, no. 9, Sep. 2018, Art. no. e0203261.
- [33] D. Tian and V. C. M. Leung, "Analysis of broadcasting delays in vehicular ad hoc networks," *Wireless Commun. Mobile Comput.*, vol. 11, pp. 1433–1445, Nov. 2011.
- [34] H. Chour, Y. Nasser, H. Artail, A. Kachouh, and A. Al-Dubai, "VANET aided D2D discovery: Delay analysis and performance," *IEEE Trans. Veh. Technol.*, vol. 66, no. 9, pp. 8059–8071, Sep. 2017.
- [35] E. Spaho, M. Ikeda, L. Barolli, F. Xhafa, A. Biberaj, and J. Iwashige, "Performance comparison of DSDV and DYMO protocols for vehicular ad hoc networks," in *Proc. IEEE 26th Int. Conf. Adv. Inf. Netw. Appl.*, Mar. 2012, pp. 629–634.
- [36] K. Xu, D. Tipper, Y. Qian, and P. Krishnamurthy, "Time-dependent performance analysis of IEEE 802.11p vehicular networks," *IEEE Trans. Veh. Technol.*, vol. 65, no. 7, pp. 5637–5651, Jul. 2016.
- [37] X. Ma, J. Zhang, and T. Wu, "Reliability analysis of one-hop safetycritical broadcast services in VANETs," *IEEE Trans. Veh. Technol.*, vol. 60, no. 8, pp. 3933–3946, Oct. 2011.
- [38] Y. P. Fallah, C.-L. Huang, R. Sengupta, and H. Krishnan, "Analysis of information dissemination in vehicular ad-hoc networks with application to cooperative vehicle safety systems," *IEEE Trans. Veh. Technol.*, vol. 60, no. 1, pp. 233–247, Jan. 2011.
- [39] Y. Mylonas, M. Lestas, A. Pitsillides, P. Ioannou, and V. Papadopoulou, "Speed adaptive probabilistic flooding for vehicular ad hoc networks," *IEEE Trans. Veh. Technol.*, vol. 64, no. 5, pp. 1973–1990, May 2015.
- [40] H. H. Saleh and S. T. Hosoon, "A survey on VANETs: Challenges and solutions," *Int. J. Eng. Technol.*, vol. 8, no. 4, pp. 266–274, Jan. 2019.
- [41] K. Jia, Y. Hou, K. Niu, C. Dong, and Z. He, "The delay-constraint broadcast combined with resource reservation mechanism and field test in VANET," *IEEE Access*, vol. 7, pp. 59600–59612, 2019.
- [42] V. Diez Rodriguez, J. Detournay, A. Vinel, and N. Lyamin, "An approach for receiver-side awareness control in vehicular ad hoc networks," *IEEE Trans. Intell. Transp. Syst.*, vol. 19, no. 4, pp. 1227–1236, Apr. 2018.
- [43] X. Zhang, L. Yan, H. Zhang, and D. K. Sung, "A concurrent transmission based broadcast scheme for urban VANETs," *IEEE Trans. Mobile Comput.*, vol. 18, no. 1, pp. 1–12, Jan. 2019.
- [44] A. Husain, R. S. Raw, B. Kumar, and A. Doegar, "Performance comparison of topology and position based routing protocols in vehicular network environments," *Int. J. Wireless Mobile Netw.*, vol. 3, pp. 289–303, Aug. 2011.
- [45] C. Harsch, A. Festag, and P. Papadimitratos, "Secure position-based routing for VANETs," in *Proc. IEEE 66th Veh. Technol. Conf.*, Sep./Oct. 2007, pp. 26–30.
- [46] J. Shao, X. Lin, R. Lu, and C. Zuo, "A threshold anonymous authentication protocol for VANETs," *IEEE Trans. Veh. Technol.*, vol. 65, no. 3, pp. 1711–1720, Mar. 2016.
- [47] A. Rostami, B. Cheng, G. Bansal, K. Sjöberg, M. Gruteser, and J. B. Kenney, "Stability challenges and enhancements for vehicular channel congestion control approaches," *IEEE Trans. Intell. Transp. Syst.*, vol. 17, no. 10, pp. 2935–2948, Oct. 2016.
- [48] M. Sepulcre, J. Gozalvez, J. Harri, and H. Hartenstein, "Applicationbased congestion control policy for the communication channel in VANETs," *IEEE Commun. Lett.*, vol. 14, no. 10, pp. 951–953, Oct. 2010.
- [49] W. Farooq, M. A. Khan, S. Rehman, and N. A. Saqib, "A survey of multicast routing protocols for vehicular ad hoc networks," *Int. J. Distrib. Sensor Netw.*, vol. 11, no. 8, 2015, Art. no. 923086.

- [50] Y. An and X. Luo, "MACRE: A novel distributed congestion control algorithm in DTN," in *Emerging Engineering Approaches and Applications* (Advanced Engineering Forum), vol. 1. Switzerland, Trans Tech Publications, 2011, pp. 71–75.
- [51] J. Sospeter, D. Wu, S. Hussain, and T. Tesfa, "An effective and efficient adaptive probability data dissemination protocol in VANET," *Data*, vol. 4, p. 1, Dec. 2018.
- [52] X. Shen, X. Cheng, R. Zhang, B. Jiao, and Y. Yang, "Distributed congestion control approaches for the IEEE 802.11p vehicular networks," *IEEE Intell. Transp. Syst. Mag.*, vol. 5, no. 4, pp. 50–61, Oct. 2013.
- [53] S. A. A. Shah, E. Ahmed, F. Xia, A. Karim, M. A. Qureshi, I. Ali, and R. M. D. Noor, "Coverage differentiation based adaptive tx-power for congestion and awareness control in VANETs," *Mobile Netw. Appl.*, vol. 23, no. 5, pp. 1194–1205, 2018.
- [54] M. Y. Taleb, S. Merniz, and S. Harous, "Congestion control techniques in VANETs: A survey," in *Proc. 13th Int. Wireless Commun. Mobile Comput. Conf. (IWCMC)*, Jun. 2017, pp. 484–488.
- [55] J. Huang, Q. Li, S. Zhong, L. Liu, P. Zhong, J. Wang, and J. Ye, "Synthesizing existing CSMA and TDMA based MAC protocols for VANETs," *Sensors*, vol. 17, no. 2, p. 338, 2017.
- [56] S. Singh, S. Negi, S. K. Verma, and N. Panwar, "Comparative study of existing data scheduling approaches and role of cloud in VANET environment," *Proc. Comput. Sci.*, vol. 125, no. 2, pp. 925–934, 2018.
- [57] F. Goudarzi and H. Asgari, "Non-cooperative beacon power control for VANETs," *IEEE Trans. Intell. Transp. Syst.*, vol. 20, no. 2, pp. 777–782, Feb. 2019.
- [58] S. A. A. Shah, E. Ahmed, J. J. P. C. Rodrigues, I. Ali, and R. M. Noor, "Shapely value perspective on adapting transmit power for periodic vehicular communications," *IEEE Trans. Intell. Transp. Syst.*, vol. 19, no. 3, pp. 977–986, Mar. 2018.
- [59] D. B. Rawat, D. C. Popescu, G. Yan, and S. Olariu, "Enhancing VANET performance by joint adaptation of transmission power and contention window size," *IEEE Trans. Parallel Distrib. Syst.*, vol. 22, no. 9, pp. 1528–1535, Jan. 2011.
- [60] M. Joseph, X. Liu, and A. Jaekel, "An adaptive power level control algorithm for DSRC congestion control," in *Proc. 8th ACM Symp. Design Anal. Intell. Veh. Netw. Appl.*, 2018, pp. 57–62.
- [61] O. M. Akinkade, "Adaptive transmission power with vehicle density for congestion control," Electronic thesis and dissertation, Univ. Windsor, Windsor, ON, Canada, Tech. Rep. 7420, 2018.
- [62] X. Liu and A. Jaekel, "Congestion control in V2V safety communication: Problem, analysis, approaches," *Electronics*, vol. 8, no. 5, p. 540, 2019.
- [63] M. Sepulcre, J. Mittag, P. Santi, H. Hartenstein, and J. Gozalvez, "Congestion and awareness control in cooperative vehicular systems," *Proc. IEEE*, vol. 99, no. 7, pp. 1260–1279, Jul. 2011.
- [64] M. Amoozadeh, H. Deng, C.-N. Chuah, H. M. Zhang, and D. Ghosal, "Platoon management with cooperative adaptive cruise control enabled by VANET," *Veh. Commun.*, vol. 2, no. 2, pp. 110–123, 2015.
- [65] E. A. Feukeu and T. Zuva, "Dynamic broadcast storm mitigation approach for VANETs," *Future Gener. Comput. Syst.*, to be published.
- [66] G. Boquet, J. L. Vicario, A. Correa, A. Morell, I. Rashdan, E. M. Díaz, and F. de Ponte Müller, "Trajectory prediction to avoid channel congestion in V2I communications," in *Proc. IEEE 28th Annu. Int. Symp. Pers., Indoor, Mobile Radio Commun. (PIMRC)*, Oct. 2017, pp. 1–6.
- [67] D. Jiang, Q. Chen, and L. Delgrossi, "Optimal data rate selection for vehicle safety communications," in *Proc. 5th ACM Int. Workshop VehiculAr Inter-NETworking (VANET)*. New York, NY, USA: ACM, 2008, pp. 30–38.
- [68] M. Sepulcre, J. Gozalvez, J. Harri, and H. Hartenstein, "Contextual communications congestion control for cooperative vehicular networks," *IEEE Trans. Wireless Commun.*, vol. 10, no. 2, pp. 385–389, Feb. 2011.
- [69] B. Aygun, M. Boban, and A. M. Wyglinski, "ECPR: Environment-and context-aware combined power and rate distributed congestion control for vehicular communications," *CoRR*, vol. abs/1502.00054, pp. 1–37, Jan. 2015.
- [70] R. Baldessari, D. Scanferla, L. Le, W. Zhang, and A. Festag, "Joining forces for VANETs: A combined transmit power and rate control algorithm," in *Proc. 7th Int. Workshop Intell. Transp.*, Mar. 2010, pp. 1–5.
- [71] S. Bolufé, S. Montejo-Sánchez, C. A. Azurdia-Meza, S. Céspedes, R. D. Souza, and E. M. G. Fernandez, "Dynamic control of beacon transmission rate and power with position error constraint in cooperative vehicular networks," in *Proc. 33rd Annu. ACM Symp. Appl. Comput.*, 2018, pp. 2084–2091.

- [72] T. Tielert, D. Jiang, H. Hartenstein, and L. Delgrossi, "Joint power/rate congestion control optimizing packet reception in vehicle safety communications," in *Proc. 10th ACM Int. Workshop Veh. Inter-Netw., Syst., Appl. (VANET).* New York, NY, USA: ACM, 2013, pp. 51–60.
- [73] V. Nguyen, C. Pham, T. Oo, N. Tran, E.-N. Huh, and C. S. Hong, "MAC protocols with dynamic interval schemes for VANETs," *Veh. Commun.*, vol. 15, pp. 40–62, Jan. 2018.
- [74] A. N. Dharsandiya and R. M. Patel, "A review on MAC protocols of vehicular ad hoc networks," in *Proc. Int. Conf. Wireless Commun., Signal Process. Netw. (WiSPNET)*, Mar. 2016, pp. 1040–1045.
- [75] M. Hadded, P. Muhlethaler, A. Laouiti, R. Zagrouba, and L. A. Saidane, "TDMA-based MAC protocols for vehicular ad hoc networks: A survey, qualitative analysis, and open research issues," *IEEE Commun. Surveys Tuts.*, vol. 17, no. 4, pp. 2461–2492, 4th Quart., 2015.
- [76] Q. Wang, S. Leng, H. Fu, and Y. Zhang, "An IEEE 802.11p-based multichannel MAC scheme with channel coordination for vehicular ad hoc networks," *IEEE Trans. Intell. Transp. Syst.*, vol. 13, no. 2, pp. 449–458, Jun. 2012.
- [77] R. Stanica, E. Chaput, and A.-L. Beylot, "Enhancements of IEEE 802.11p protocol for access control on a VANET control channel," in *Proc. IEEE Int. Conf. Commun. (ICC)*, Jun. 2011, pp. 1–5.
- [78] R. Stanica, E. Chaput, and A.-L. Beylot, "Congestion control in csmabased vehicular networks: Do not forget the carrier sensing," in *Proc.* 9th Annu. IEEE Commun. Soc. Conf. Sensor, Mesh Ad Hoc Commun. Netw. (SECON), Jun. 2012, pp. 650–658.
- [79] Y. Lu, J. Ren, J. Qian, M. Han, Y. Huo, and T. Jing, "Predictive contention window-based broadcast collision mitigation strategy for VANET," in Proc. IEEE Int. Conf. Big Data Cloud Comput. (BDCloud), Social Comput. Netw. (SocialCom), Sustain. Comput. Commun. (SustainCom) (BDCloud-SocialCom-SustainCom), Oct. 2016, pp. 209–215.
- [80] C. Cooper, D. Franklin, M. Ros, F. Safaei, and M. Abolhasan, "A comparative survey of VANET clustering techniques," *IEEE Commun. Surveys Tuts.*, vol. 19, no. 1, pp. 657–681, 1st Quart., 2017.
- [81] G. H. Alsuhli, A. Khattab, and Y. A. Fahmy, "Double-head clustering for resilient VANETs," *Wireless Commun. Mobile Comput.*, vol. 2019, Mar. 2019, Art. no. 2917238.
- [82] M. Ren, J. Zhang, L. Khoukhi, H. Labiod, and V. Vèque, "A unified framework of clustering approach in vehicular ad hoc networks," *IEEE Trans. Intell. Transp. Syst.*, vol. 19, no. 5, pp. 1401–1414, May 2018.
- [83] L. Liu, C. Chen, T. Qiu, M. Zhang, S. Li, and B. Zhou, "A data dissemination scheme based on clustering and probabilistic broadcasting in VANETs," *Veh. Commun.*, vol. 13, pp. 78–88, Jul. 2018.
- [84] S. Ucar, S. C. Ergen, and O. Ozkasap, "Multihop-Cluster-Based IEEE 802.11p and LTE hybrid architecture for VANET safety message dissemination," *IEEE Trans. Veh. Technol.*, vol. 65, no. 4, pp. 2621–2636, Apr. 2015.
- [85] M. B. Taha, C. Talhi, and H. Ould-Slimanec, "A cluster of CP-ABE microservices for VANET," *Proc. Comput. Sci.*, vol. 155, pp. 441–448, Aug. 2019.
- [86] G. Guo and S. Wen, "Communication scheduling and control of a platoon of vehicles in VANETs," *IEEE Trans. Intell. Transp. Syst.*, vol. 17, no. 6, pp. 1551–1563, Jun. 2016.
- [87] J. Choi and Y. Yi, "Joint optimization of message transmissions with adaptive selection of CCH interval in VANETs," *IEEE Access*, vol. 7, pp. 72090–72104, 2019.
- [88] M. Y. B. Darus and K. A. Bakar, "Congestion control algorithm for disseminating Uni-priority safety messages in VANETS," *Int. J. Digit. Content Technol. Appl.*, vol. 7, no. 7, p. 41, 2013.
- [89] K. Nellore and G. P. Hancke, "Traffic management for emergency vehicle priority based on visual sensing," *Sensors*, vol. 16, no. 11, p. 1892, 2016.
- [90] N. Noorani and S. A. H. Seno, "Routing in VANETs based on intersection using SDN and fog computing," in *Proc. 8th Int. Conf. Comput. Knowl. Eng. (ICCKE)*, Oct. 2018, pp. 339–344.
- [91] W. Qi, Q. Song, X. Wang, L. Guo, and Z. Ning, "SDN-enabled social-aware clustering in 5G-VANET systems," *IEEE Access*, vol. 6, pp. 28213–28224, 2018.
- [92] S. A. Hussain and A. Saeed, "An analysis of simulators for vehicular ad hoc networks," *World Applied Sciences Journal*, vol. 23, no. 8, pp. 1044–1048, 2013.
- [93] D. Krajzewicz, J. Erdmann, M. Behrisch, and L. Bieker, "Recent development and applications of sumo-simulation of urban mobility," *Int. J. Adv. Syst. Meas.*, vol. 5, nos. 3–4, pp. 128–138, 2012.

- [94] K.-C. Lan, "MOVE: A practical simulator for mobility model in VANET," in *Telematics Communication Technologies and Vehicular Networks: Wireless Architectures and Applications*. Washington, DC, USA: IGI Global, 2010, pp. 355–368.
- [95] D. R. Choffnes and F. E. Bustamante, "An integrated mobility and traffic model for vehicular wireless networks," in *Proc. 2nd ACM Int. Workshop Veh. Ad Hoc Netw. (VANET).* New York, NY, USA: ACM, 2005, pp. 69–78.
- [96] J. Härri, M. Fiore, F. Filali, and C. Bonnet, "Vehicular mobility simulation with VANETMobiSim," *Simulation*, vol. 87, no. 4, pp. 275–300, 2009.
- [97] G. D. B. Cameron and G. I. D. Duncan, "PARAMICS—Parallel microscopic simulation of road traffic," J. Supercomput., vol. 10, no. 1, pp. 25–53, Mar. 1996.
- [98] M. Güneş, F. Juraschek, B. Blywis, and C. Graff, "MoNoTrac—A mobility trace generator based on OpenStreetMap geo-data," in *Proc. 7th IEEE Int. Conf. Mobile Ad-Hoc Sensor Syst. (IEEE MASS)*, Nov. 2010, pp. 618–623.
- [99] T. Issariyakul and E. Hossain, "Introduction to network simulator 2 (NS2)," in *Introduction to Network Simulator NS2*. Switzerland: Springer, 2009, pp. 1–18.
- [100] T. R. Henderson, M. Lacage, and G. F. Riley, "Network simulations with the ns-3 simulator," *SIGCOMM Demonstration*, vol. 14, no. 14, p. 527, 2008.
- [101] A. Varga and R. Hornig, "An overview of the OMNeT++ simulation environment," in *Proc. 1st Int. Conf. Simul. Tools Techn. Commun.*, *Netw. Syst. Workshops (ICST)* (Institute for Computer Sciences, Social-Informatics and Telecommunications Engineering). Brussels, Belgium: ICST, 2008, pp. 60:1–60:10.
- [102] B. Ahmed, A. W. Malik, T. Hafeez, and N. Ahmed, "Services and simulation frameworks for vehicular cloud computing: A contemporary survey," *EURASIP J. Wireless Commun. Netw.*, vol. 2019, no. 1, p. 4, Jan. 2019.
- [103] X. Chang, "Network simulations with OPNET," in Proc. Winter Simulation Conf., Simulation—A Bridge Future, vol. 1. New York, NY, USA: ACM, 1999, pp. 307–314.
- [104] S.-Y. Wang, C.-L. Chou, and C.-M. Yang, "EstiNet openflow network simulator and emulator," *IEEE Commun. Mag.*, vol. 51, no. 9, pp. 110–117, Sep. 2013.
- [105] C. Sommer, R. German, and F. Dressler, "Bidirectionally coupled network and road traffic simulation for improved IVC analysis," *IEEE Trans. Mobile Comput.*, vol. 10, no. 1, pp. 3–15, Jan. 2011.
- [106] R. Protzmann, B. Schünemann, and I. Radusch, "Simulation of convergent networks for intelligent transport systems with VSimRTI," in *Networking Simulation for Intelligent Transportation Systems: High Mobile Wireless Nodes.* Hoboken, NJ, USA: Wiley, 2017, pp. 1–28.
- [107] A. Zarrad and I. Alsmadi, "Evaluating network test scenarios for network simulators systems," *Int. J. Distrib. Sensor Netw.*, vol. 13, no. 10, 2017, Art. no. 1550147717738216.



**BALAWAL SHABIR** received the bachelor's degree in electrical engineering and the master's degree in information security from Air University Islamabad. He is currently pursuing the Ph.D. degree with NUST, Pakistan. His research interests include cryptography and network security, privacy, vehicular adhoc networks, the Internet of Things, big data analytics, machine learning, bioinformatics, and cloud computing.



**MUAZZAM A. KHAN** received the Ph.D. degree in computer science, as a joint program, from The International Islamic University, Islamabad (IIUI) and The University of Missouri–Kansas City (UMKC), USA, in 2011. He held a postdoctoral position at the University of Ulm, in 2013, and the University of Missouri, in 2016. He had worked at the Networking and Multimedia Lab, UMKC, as a Research Fellow. He is currently working as a Tenured Associate Professor/Associate Dean at

SEECS, National University of Sciences and Technology (NUST), Pakistan. His research interests include wireless sensor networks, body area networks, image compression, image encryption, and data network security.



ANIS U. RAHMAN received the master's degree in parallel and distributed systems from Joseph Fourier University, France, and the Ph.D. degree in computer science from Grenoble University, France, in 2013. He is currently an Assistant Professor with NUST-SEECS, Pakistan. Besides, he is also working as Research Fellow of the Faculty of Computer Science and Information Technology, University of Malaya, Malaysia. His main research interests include the Internet of

Things and machine learning.



**ASAD W. MALIK** received the Ph.D. degree, with majors in parallel and distributed simulation/systems, from NUST, Pakistan, in 2012. He is also working as a Senior Lecturer with the Department of Information Systems, Faculty of Computer Science and Information Technology, University of Malaya, Malaysia. He is currently an Assistant Professor with NUST-SEECS, Pakistan. His primary area of interests include distributed simulation, cloud/fog computing, and the Internet of Things.



**ABDUL WAHID** received the Ph.D. degree from Kyungpook National University, South Korea. He is currently an Assistant Professor with NUST-SEECS, Pakistan. His research interests include vehicular ad hoc network, wireless sensor networks, underwater wireless sensor networks, cyber-physical systems, software-defined networking, and information-centric networking. He is also a Reviewer and a TPC member of many conferences and journals.

• • •